

MIT Open Access Articles

Satellite#Based Assessment of Land Surface Energy Partitioning– Soil Moisture Relationships and Effects of Confounding Variables

The MIT Faculty has made this article openly available. *Please share* how this access benefits you. Your story matters.

Citation: Feldman, Andrew F. et al., "Satellite#Based Assessment of Land Surface Energy Partitioning–Soil Moisture Relationships and Effects of Confounding Variables." Water Resources Research 55, 12 (December 2019): 10657-77 doi. 10.1029/2019WR025874 ©2019 Authors

As Published: https://dx.doi.org/10.1029/2019WR025874

Publisher: American Geophysical Union (AGU)

Persistent URL: https://hdl.handle.net/1721.1/125701

Version: Final published version: final published article, as it appeared in a journal, conference proceedings, or other formally published context

Terms of use: Creative Commons Attribution 4.0 International license





Water Resources Research

RESEARCH ARTICLE

10.1029/2019WR025874

Key Points:

- Soil moisture threshold separating water and energy-limited evaporation regimes is estimated across diverse African biomes
- Coupling between moisture and energetic partitioning under water limitation is strongest in drier African biomes
- Anomalous vapor pressure deficit time evolution during surface drying can confound energetic partitioning--soil moisture relationships

Supporting Information:

• Supporting Information S1

Correspondence to:

A. F. Feldman, afeld24@mit.edu

Citation:

Feldman, A. F., Short Gianotti, D. J., Trigo, I. F., Salvucci, G. D., & Entekhabi, D. (2019). Satellite-based assessment of land surface energy partitioning-soil moisture relationships and effects of confounding variables. *Water Resources Research*, *55*, 10,657–10,677. https://doi.org/10.1029/ 2019WR025874

Received 27 JUN 2019 Accepted 22 NOV 2019 Accepted article online 11 DEC 2019 Published online 12 DEC 2019

©2019. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Satellite-Based Assessment of Land Surface Energy Partitioning–Soil Moisture Relationships and Effects of Confounding Variables

Andrew F. Feldman¹, Daniel J. Short Gianotti¹, Isabel F. Trigo^{2,3}, Guido D. Salvucci⁴, and Dara Entekhabi¹

¹Department of Civil and Environmental Engineering, Massachusetts Institute of Technology, Cambridge, MA, USA,
 ²Instituto Português do Mar e da Atmosfera I.P. (IPMA), Lisbon, Portugal, ³Instituto Dom Luiz (IDL), Lisbon, Portugal,
 ⁴Department of Earth and Environment, Boston University, Boston, MA, USA

Abstract Land surface energetic partitioning between latent, sensible, and ground heat fluxes determines climate and influences the terrestrial segment of land-atmosphere coupling. Soil moisture, among other variables, has a direct influence on this partitioning. Dry surfaces characterize a water-limited regime where evapotranspiration and soil moisture are coupled. This coupling is subdued for wet surfaces, or an energy-limited regime. This framework is commonly evaluated using the evaporative fraction--soil moisture relationship. However, this relationship is explicitly or implicitly prescribed in land surface models. These impositions, in turn, confound model-based evaluations of energetic partitioning--soil moisture relationships. In this study, we use satellite-based observations of surface temperature diurnal amplitude (directly related to available energy partitioning) and soil moisture, free of model impositions, to estimate characteristics of surface energetic partitioning--soil moisture relationships during 10--20-day surface drying periods across Africa. We specifically estimate the spatial patterns of water-limited energy flux sensitivity to soil moisture (m) and the soil moisture threshold separating water and energy-limited regimes (θ^*). We also assess how time evolution of other factors (e.g., solar radiation, vapor pressure deficit, surface albedo, and wind speed) can confound the energetic partitioning--soil moisture relationship. We find higher m in drier regions and interestingly similar spatial θ^* distributions across biomes. Vapor pressure deficit and insolation increases during drying tend to increase m. Only vapor pressure deficit increases in the Sahelian grasslands systematically decrease θ^* . Ultimately, soil and atmospheric moisture availability together play the largest role in land surface energy partitioning with minimal consistent influences of time evolution of other forcings.

Plain Language Summary Whether available, incoming solar energy is used for evaporating surface water, or surface heating largely depends on water availability across the landscape. Under dry conditions (water limitation), increasing soil moisture increases evaporation and surface cooling. In this regime, droughts and heatwaves can be initiated and sustained because drying is positively reinforced. Under wetter conditions (energy limitation), increasing soil moisture does not generally influence evaporation. Climate models rely on these soil moisture-evaporation relationships to describe associations between water and energy cycles and predict future climate. However, due to difficulty observing evaporation at large scales, these relationships assume different forms across climate models which contribute to divergences and uncertainty in making climate projections. We use satellite observations of soil moisture and daily temperature range (quantifying surface heating; inversely related to evaporation) to evaluate these relationships free of model impositions across Africa. Following rain events during surface drying, daily temperature range increases with greatest increases in drylands indicating strong surface influence on atmosphere. Increasing incoming sunlight and atmospheric dryness after rainfall accompany and amplify this surface heating. Regions with surface moisture strongly influencing surface heating/evaporation, such as the Sahel, are expected to have soil moisture variations that can initiate future rain and/or drought/heatwave events.

1. Introduction

The water, carbon, and energy cycles are coupled at the land surface principally through the soil moisture and soil temperature state variables. These state variables can also restore perturbations or prolong them. For example, soil moisture modulated land-atmosphere coupling has implications for feedbacks that drive



Figure 1. Schematic of the typical relationship between evaporative fraction and soil moisture. A third dimension indicates that aspects of this relationship could change with other factors. ET=evapotranspiration; Rs= insolation; VPD=vapor pressure deficit; ∂ =surface albedo; θ =soil moisture.

rainfall or the lack thereof and can lead to persistence of droughts and heatwaves (Miralles et al., 2019; Rodriguez-Iturbe et al., 1991; Santanello et al., 2018). Under water-limited conditions, soil moisture (θ) directly controls evapotranspiration (ET) and surface temperature, driving a positive land-atmosphere feedback loop (Seneviratne et al., 2010). In this regime, higher θ increases ET which is coupled with greater near surface humidity and a reduction in surface temperature, cloud base, and dry air entrainment into the boundary layer (Betts, 2004; Betts et al., 1996; Entekhabi et al., 1996; Seneviratne et al., 2010). There is general agreement that these conditions can increase the probability of rainfall keeping the surface wet (Findell et al., 1997; Findell & Eltahir, 2003; Koster et al., 2004; Tuttle & Salvucci, 2016), though caveats of convective triggering under dry surface anomalies are noted (Findell & Eltahir, 2003; Taylor et al., 2012, 2011). Conversely, a drying surface without moisture replenishment leads to less evaporative cooling. Corresponding increases in surface albedo further inhibit vertical motion and rainfall (Charney, 1975), and the lower atmosphere ultimately dries and warms. These feedbacks lead to persistence of drought and heatwaves (Hirschi et al., 2011; Miralles et al., 2019; Roundy et al., 2012), until the system is sufficiently shocked by exogenous (generally advective) restoring forces (Brubaker & Entekhabi, 1996). At higher mean θ availability, the system is considered energy-limited as more moisture does not necessarily lead to greater ET. Here, the water, carbon, and energy cycle couplings are subdued. ET is at or near its potential value where net radiation and atmospheric resistance, for example, are instead limiting.

These coupled processes are typically evaluated using the relationship between evaporative fraction (EF) and θ where EF is

$$EF = \frac{LE}{LE + H} = \frac{LE}{R_n - G},\tag{1}$$

$$R_n = LE + H + G,\tag{2}$$

where LE is latent heat flux, H is sensible heat flux, G is ground heat flux, and R_n is net radiation. The available net radiative energy at the surface can be dissipated via LE, H, and G in equation (2). All flux terms have positive signs when away from the surface. EF is therefore the fraction of available surface energy partitioned into evaporating water (LE) instead of heating the overlying air (H). The canonical EF- θ relationship is characterized by a transition point in soil moisture (θ^* ; some other studies label θ_{Crit} ; Seneviratne et al., 2010) separating the water (Stage II ET) and energy-limited (Stage I ET) regimes (Budyko, 1974; Eagleson, 1978). A conceptual drawing of this function is shown in Figure 1. Below θ^* , EF and θ are positively related

(Teuling et al., 2006) where reduced θ lowers EF. The relationship may be nonlinear, but the first-order linear term of the function across this segment below θ^* is indicative of the degree of coupling between the surface water and energy balances. Above θ^* , EF is decoupled from soil moisture, and hence, its functional relationship is a flat line with an approximately zero slope. Here, EF is an influenced by other variables such as net radiation (Gentine et al., 2007).

 θ^* itself is largely defined by soil texture and plant physiology. Specifically, θ^* defines the θ value below which soil water controls the combined rates of bare soil evaporation and water loss through stomata (Eagleson, 1978; Seneviratne et al., 2010). In other words, ET drops below its potential or energy-limited value. At a given location, the probability density function of θ together with θ^* estimate describes the time spent in each hydrologic regime, key for determining the existence of land-atmosphere feedbacks. While θ^* divides the system into water and energy-limited states of the system, $\frac{\partial EF}{\partial A}$ (the sensitivity of EF to θ below θ^*) determines the magnitude of θ control on the boundary layer (Dirmeyer, 2011). The local slope is only a potential metric of influence, however, as θ variations must also be of sufficient magnitude to lead to variability in boundary layer processes (Dirmeyer, 2011; Guo et al., 2006). Together, both bare soil and vegetation components contribute to the functional behavior below θ^* . For bare soils, greater θ leads to less soil capillary pressure retention and more efficient atmospheric extraction of soil vapor. For vegetated surfaces, greater θ is associated with plants keeping stomatal apertures open to transpire and assimilate carbon. It is thus no surprise that the general relationship between EF and θ in Figure 1 is also observed between primary productivity and θ , evidence for coupling to the carbon cycle (Short Gianotti et al., 2019). Ultimately, accurate representation of surface energy partitioning in land surface models is essential for weather forecasting and climate projections of ET and gross primary productivity.

1.1. Surface Energy Partitioning-Soil Moisture Relationship: Model Assessments

Energetic partitioning– θ relationships remain largely unobserved due to the challenge of directly measuring surface energy fluxes and θ at climate model grid scales across different climates and biomes. These relationships have primarily been evaluated and visualized via EF– θ within model frameworks (Dirmeyer et al., 2000; Koster et al., 2009; Schwingshackl et al., 2017). These studies revealed much about vegetation dependencies, inter-model differences, and land-atmosphere interactions. However, any model-based evaluation of EF– θ is subject to a combination of explicit and/or implicit impositions that ultimately formulate their relationship (Dirmeyer et al., 2007; Koster et al., 2006). Any parameterization that translates even a purely observed θ to an ET estimate is based on land surface energy partitioning assumptions that impose θ^* and $\frac{\partial EF}{\partial \theta}$, for instance. Model-free observations of this relationship are required to fully evaluate climate and vegetation dependencies and are critical for diagnosing land-atmosphere feedbacks in models (Dirmeyer et al., 2007; Guo et al., 2006). Observed temperature–precipitation relationships (with measurements of each gridded at large spatial scales) were used to compare with or serve as a proxy for EF– θ relationships (Gevaert et al., 2018; Koster et al., 2009). However, differences between EF and temperature and precipitation and θ hinder one-to-one comparison between these relationships.

1.2. Surface Energy Partitioning-Soil Moisture Relationship: Observations

The most direct observations of energetic partitioning can be obtained from flux towers (e.g., using eddy covariance measurements of latent and sensible heat fluxes), though with known errors associated with scaling measurements up to a land surface model grid (Crow & Wood, 2002; Teuling et al., 2006). Also, the flux towers do not adequately cover the full range of combinations of climates and biomes. The newly available θ satellite remote sensing products, especially based on lower microwave frequencies (L-band) with reduced vegetation canopy opacity and greater moisture sensing depth (Entekhabi et al., 2010; Kerr et al., 2010), allow more direct observations of surface energy partitioning– θ relationships at spatial scales of land surface models. These remote sensing θ observations, when used alongside surface temperature metrics, have been shown to provide robust climate coupling assessments (Dong & Crow, 2018; Dong & Crow, 2019). Time series of satellite-derived θ during inter-storm periods (Akbar et al., 2018) indicate that surface drying holds information about ET and hydrologic regimes comparable to those from models (Koster et al., 2009). Satellite θ combined with station-based observations of EF revealed that these EF– θ relationships hold temporally at a given location over seasonal timescales as well as spatially using mean fields (Short Gianotti et al., 2019). Climate and vegetation constraints on surface energy partitioning are apparent from these studies. However, fundamental characteristics of the surface energy partitioning– θ relationships outlined in Figure 1 remain to be quantified from observations.

We do not use EF here as no direct, model-free estimations of ET exist at the land surface model grid scale. Instead, information from the diurnal evolution of land surface temperature is used here as it bears the signature of surface energy partitioning and can be observed using satellite remote sensing (Bateni & Entekhabi, 2012; Panwar et al., 2019). Specifically, increasing surface temperature diurnal amplitude (dT) in the water-limited regime can largely reflect the surface sensible heat flux becoming more efficient than latent heat flux at dissipating surface energy (Panwar et al., 2019). This manifests as less evaporative cooling from morning to afternoon at the land surface. dT is also expected to exhibit a lower magnitude of coupling with θ at high θ (under energy limitation). The θ -dependent division of behavior between high and low θ , if estimated robustly, is θ^* . Positive coupling between dT and sensible heating and their negative coupling with EF and θ have been noted previously (Amano & Salvucci, 1999; Betts et al., 2014).

1.3. Surface Energy Partitioning-Soil Moisture Relationship: Potential Confounding Effects

The surface energetic partitioning– θ relationship is affected by other factors. The EF– θ relationship, for example, is a multidimensional function with dependence on factors, such as insolation (Rs), albedo (α), wind speed, and vapor pressure deficit (VPD), as confirmed by flux tower observations (Ford et al., 2014; Gu et al., 2006; Haghighi et al., 2018). The scatter in EF– θ plots is partly due to ignoring these dependencies and projecting EF only onto θ space. This can manifest as shifts in θ^* and $\frac{\partial EF}{\partial \theta}$ as shown by changes from Curves 1 to 2 in Figure 1. This framework was investigated for bare surfaces with findings that θ^* can change significantly with temperature and humidity (Haghighi et al., 2018). The θ^* threshold defines the θ value at which ET equals or sufficiently approaches potential ET. When considering widely used ET theory, this breakpoint depends on when the ratio of surface to atmospheric resistance approaches a small value (Peng et al., 2019). This condition can be seen when setting Penman potential ET and Penman–Monteith actual ET equal, for instance. This ratio is a function of all aforementioned factors: VPD, wind speed, Rs, α , and θ . While theoretical formulations like Penman–Monteith do not always match observations, it is conceivable that systematic changes in these variables to varying degrees can translate to different θ^* . In the pursuit of detecting θ^* and Stage II energetic partitioning sensitivity to θ , it is important to quantify and isolate the effects of the aforementioned potential confounding factors.

1.4. Study Goals

In this study, we use satellite observations of surface θ and dT with near daily sampling to evaluate the relationship between land surface energy partitioning and θ across a range of natural biomes in Africa. The θ fields are based on a low-Earth orbit satellite with a low-frequency microwave radiometer and global coverage. The dT range fields require resolving the diurnal cycle of the variable and hence require geostationary satellite records. For this, we use a multichannel optical to infrared sensor suite in geostationary orbit with full coverage of Africa (nadir of disc is centered on the Equator in Africa). We ask: (1) across climates and biomes, what are model-free estimates of (a) θ -dependent critical points between water and energy-limited regimes (θ^*) and (b) surface energetic partitioning sensitivity to θ in the water-limited regime. (2) Where are frequent transitions between hydrologic regimes occurring? (3) What is the temporal evolution of potential confounding factors during transitions from energy to water-limited regimes, and how might their dynamics confound surface energy partitioning– θ relationships at large spatial scales? Our objectives are thus to observe surface energy partitioning– θ relationships at land surface model scales, identify locations of hydrologic regime transitions, and assess whether variability of these relationships as found at tower sites can be consistently explained by other factors.

2. Methods

2.1. Soil Moisture Drydown Framework

The surface energetic partitioning– θ relationship has been shown to have strong climate/vegetation controls with spatial mean patterns and averaged temporal patterns (monthly) at a given location (Koster et al., 2009; Short Gianotti et al., 2019). In this study, we instead investigate these relationships during drying periods following rainfall where θ abruptly rises then decays until the next rain event, referred to henceforth as drydowns. A canonical drydown is shown in Figure 2a where the surface begins in an energy-limited state and





Figure 2. (a) Soil moisture drydown framework used to assess observed evaporative fraction-soil moisture curves in this study. The dashed line indicates when θ^* and thus transition to water limitation is reached. The rates of change of each variable are arbitrary. (b) FLUXNET Tower observations (Tier 1) during drying from 2000-2010 in a savanna in South Africa (Za-Kru; DOI: 10.18140/FLX/1440188). Note that all fluxes are measured and energy balance does not fully close. Values aggregated to medians in soil moisture bins. See Section 2.4.1 for drydown identification. dT=diurnal temperature; G=ground heat flux; H=sensible heat flux; LE=latent heat flux.

decays below θ^* into a water-limited state. We condition on these components of the time series for several reasons. (1) We can potentially isolate transitions from energy to water-limited regimes allowing more direct estimation of θ^* and water-limited energetic partitioning sensitivity to θ . (2) These shorter time periods allow evaluation of potential confounding factor dynamics that might be otherwise suppressed with temporal aggregation (i.e., monthly and annually). (3) We are removing confounding effects of ET physics due to rainfall (McColl et al., 2019), frozen periods, and completely dry periods with no θ variations (dry season). (4) We take advantage of evaluating short-term (near daily) time evolution during drying segments to remove seasonal behavior and its many confounding factors associated with the time of year (vegetation phenology, mesoscale processes, etc.) rather than direct variable interactions (Tuttle & Salvucci, 2017).

Expected behavior of dT and EF is shown in Figure 2a. The expected evolution of these variables during drying is based on previous findings for EF (Teuling et al., 2006), dT (Amano & Salvucci, 1999; Betts et al., 2013; Idso et al., 1975), surface-air temperature gradients (Gallego-Elvira et al., 2016), and surface temperature (Lakshmi et al., 2003).

We initially investigate how dT evolves in time along with surface energy fluxes during drying using eddy covariance measurements in a semiarid savanna (Figure 2b) (Scholes et al., 2001). dT and EF evolve as expected during drying. However, G, a function of both soil thermal conductivity and soil temperature gradient (Castelli et al., 1999), tends to increase along with H (Santanello & Friedl, 2003). Despite soil drying leading to lower thermal conductivity, surface soil heating during drying likely results in a significant thermal gradient driving an increasing energy flux into the surface (greater G). G is commonly not discussed as a factor in climate coupling assessments using EF–soil moisture relationships despite being potentially a significant energy flux (Figure 2b) closely tied to the surface energy balance (Gentine et al., 2007). We argue here that dT is measuring a sensible heat flux signature that has a greater magnitude than G (Figure 2b) that might be subdued by G increases (less heating of atmosphere due to available energy used for soil thermal heating). Similarly, G was previously found to be less effective at dissipating incoming energy than H and LE (Bateni & Entekhabi, 2012). Ultimately, dT is a surface energetic partitioning metric and is expected to capture the surface energy balance's transition from energy to water-limited regimes during drying regardless of which energy flux component is dominant. In other words, relative changes in LE, H, and G are all components of the dT signal, not confounding factors.

2.2. Study Region

Africa is chosen as the study region due to collocation of θ from NASA's Soil Moisture Active Passive (SMAP) satellite (Entekhabi et al., 2010) and observations from the Spinning Enhanced Visible and Infrared Imager

(SEVIRI) on-board European and EUMETSAT space agency's Meteosat Second Generation (MSG-2) geostationary satellite series (Trigo et al., 2011). This synergy of datasets allows subweekly observations of energetic partitioning– θ relationships that can resolve post rainfall period dynamics. Africa additionally includes strong climate/vegetation cover gradients for evaluating this relationships' dependence on climate and vegetation.

2.3. Datasets

2.3.1. Soil Moisture

Surface θ (approximately top 5 cm) is estimated from NASA's SMAP satellite measurements of Earth's natural upwelling emission (Entekhabi et al., 2010). This includes 4 years (1 April 2015 to 31 March 2019) of the L1C brightness temperature product at an enhanced 9-km resolution on Equal-Area Scalable Earth 2 grid using Backus Gilbert optimal interpolation (Chaubell, Chan, et al., 2016; Chaubell, Yueh, et al., 2016). The multitemporal dual channel algorithm approach is used to estimate θ which is not subject to ancillary vegetation information (Konings et al., 2016). Comparison statistics to in situ measurements are similar to those from the validated SMAP baseline θ product (Chan et al., 2016). See Feldman et al. (2018) methods for more details. The SMAP satellite samples a given location every 2 to 3 days. All further discussed datasets are linearly resampled to this grid and temporal resolution.

2.3.2. Meteorological Data

Meteorological observations are obtained from the MSG-2 geostationary satellite (Trigo et al., 2011). It measures at multiple bands within the visible and infrared spectra with a maximum spatial sampling of 3 km at the subsatellite point (nadir), which is located at 0° longitude and 0° latitude. The daily downward surface solar radiation (Carrer et al., 2011; Ceamanos et al., 2014; Geiger, Meurey, et al., 2008) and daily surface albedo (Carrer et al., 2018; Geiger, Carrer, et al., 2008) products are used for confounding factor evaluations. Additionally, land surface temperatures (LST) (Freitas et al., 2010) at 6:00 AM and 1:30 PM are obtained by averaging the 15-min measurements at a 1-hr window around these times. All MSG-2 products are provided by the EUMETSAT Satellite Applications Facility on Land Surface Analysis (Trigo et al., 2011). The 1:30 PM and 6:00 AM temperatures are then subtracted to obtain dT with only days with both values available retained. These times are representative of the temperature amplitude for this dataset in Africa (Holmes et al., 2015, 2013). Since the land surface temperature product is rigorously filtered for cloud contamination, days during rainfall are removed, replacing the need to otherwise filter out drydowns using relatively sparse precipitation datasets over Africa. This results in a large reduction in sample size (~60%). Therefore, for a given day with a SMAP overpass, any available dT observations since the previous SMAP overpass are averaged for a given pixel. This typically doubles available (θ and dT) pairs with the cost of introducing a slight mismatch between exact timing of θ and dT for a given SMAP overpass time. However, the temporal mismatch between dT and θ samples is small or 1–3 days due to SMAP θ sampling interval. Also, drydown detection (see below) intrinsically filters out cases of non-negligible storms between these 1–3-day SMAP θ sampling intervals that would introduce representativeness errors in the aggregated dT value. Alternatively, one could gap fill by linearly interpolating θ or combining θ products from other satellites (Kerr et al., 2010), though these methods are not attempted here.

Satellite LST is a directional variable since it corresponds to the radiometric temperature of the surface as seen by the sensor (Cao et al., 2019). A bias can therefore be introduced at high incidence angles where sensor view of vegetation, for example, becomes elongated. However, the LST measurement deviation from a reference view measurement (e.g., nadir view) is less than 1 to 2K for incidence angles of less than 40° (Ermida et al., 2018), the case over the majority of the this study's domain. Furthermore, the use of a geostationary platform is an advantage because the viewing geometry is time invariant (i.e., no change between drydowns), removing confounding effects from changing viewing geometry. Additionally, the geostationary orbit captures the entire LST diurnal cycle providing the most reliable dT estimates.

Wind speed and VPD, also used for confounding factor evaluations, are obtained from MERRA-2 reanalysis as no known gridded observational dataset over this region and time period exist (GMAO, 2015, 2008). Daily VPD is computed using MERRA-2 surface pressure and 2-m air temperature and specific humidity fields at 1:30 PM local time (no temporal interpolation needed as fields are reported every half hour).



Figure 3. Example model fitting procedure in example 0.5° grassland pixel in Mali (Latitude = 15.5° N; Longitude = 6.5° W). (a) Akaike Information Criterion (AIC) model selection on (θ and dT) pairs. (b) Regression approach where the AIC model selection approach chose the transitional model on (θ and dT^C) pairs. dT^C represents removing the linear effects of all covarying factors from dT. The transitional model from panel A is plotted here for reference. (c) Conditioning approach where drydowns with highest and lowest changes in vapor pressure deficit (VPD) are fit separately and shown as an example.

2.3.3. Ancillary Information

International Geosphere-Biosphere Programme land cover classifications are used to segment the results across ecohydroclimatic regions in Africa (Kim, 2013). Mean annual precipitation between 2015 and 2017 from GPM IMERG V5 is used to compare to spatial variations in metric estimates (Huffman, 2015). This ancillary information is not used in the estimation of thresholds and sensitivities.

2.4. Energetic Partitioning- θ Relationship Analysis

2.4.1. θ Drydown Identification

A similar procedure to previous studies is used to identify SMAP θ drydowns (Akbar et al., 2018; Feldman et al., 2018; McColl et al., 2017; Shellito et al., 2018). θ drydowns are identified for each 9 × 9-km² pixel as periods of consecutively decreasing θ for at least four SMAP overpasses (10–12 days) and continuing for no more than 20 days. The choice of these time periods is to filter out noisy θ periods and long drydowns that incorporate seasonal behavior. All drydowns must have at least a 0.01 m³ m⁻³ θ increase from the previous overpass to begin the drydown, suggesting occurrence of rainfall. All aforementioned meteorological variables coinciding with these times are retained for the remainder of the analysis.

Seasonal trends in each variable during drydowns are removed with the same procedure as Feldman et al. (2018). Seasonal cycles are obtained by creating a mean climatology from averaging all 4 years of the data and smoothing the series with a 30-day moving average (Dong & Crow, 2018). A tutorial of this process is shown in Figure S1. This seasonality removal approach retains the absolute magnitude of all variables as opposed to converting into anomaly metrics.

2.4.2. θ^* and *m* Estimation

 θ^* and energetic partitioning sensitivity to θ in the water-limited regime (*m*) in the context of Figure 1 are estimated by fitting a piecewise linear model to the dT- θ relationship during drydowns. To ensure adequate (θ and dT) pairs during drying periods, the analysis is conducted at a 0.5° resolution containing

approximately 36 SMAP pixels. A representative example relationship is shown in Figure 3a. Visual inspections of numerous dT– θ relationships reveal approximately piecewise linear relationships with water limitation (with high dT– θ slope) at lower θ and energy limitation (with negligible dT– θ slope) at higher θ , as in Figure 3a. Our purposes here are to estimate an identifiable breakpoint of when dT increases at lower values (θ^*) and the magnitude of dT increases per θ decrease of 0.1 m³/m³ below this breakpoint (*m*). A simple piecewise linear model was deemed adequate to estimate these metrics, even if slight curvilinearities in the data exist. These simple models have been previously used in similar frameworks (Akbar et al., 2018; Schwingshackl et al., 2017). A shortcoming is that *m* as computed from dT may not have a one-to-one equivalence with the magnitudes of EF estimated *m*. This is due to physical differences between EF and dT where, for example, warm air advection or perhaps ground heat flux will differentially affect dT more than EF. The estimated θ^* is considered a mean state value as it was established that this value is not necessarily constant (Haghighi et al., 2018). We evaluate in later sections how anomalous conditions can systematically shift θ^* .

In some regions, a θ^* and thus a transitional hydrologic regime may not be present due to θ varying only within a water or energy-limited regime. Therefore, two simpler models reflecting these respective conditions are also fit to the data in each pixel as shown in Figure 3a. Modeled dT is compared to the dT measurements with residual sum of squares (RSS) for each model. These models are equivalent to classes A to C in Akbar et al. (2018). Note that while θ^* is only estimated for the transitional model case, *m* is estimated for both water limited and transitional model cases.

To select the "best" model and avoid overfitting the data in adding parameters progressively from energylimited (two parameters), to water-limited (three parameters), to transitional regime models (four parameters), the Akaike Information Criterion (AIC) (Akaike, 1974) is used to minimize RSS and number of model parameters (k):

$$AIC = 2k + Nln(RSS), \tag{3}$$

where N is number of sample (θ and dT) pairs. The lowest AIC determines which model is selected.

Fitting a model only to the phase space of dT and θ ignores the temporal evolution of dT during drydowns. This therefore leaves the possibility open that the phase space as in Figure 3a shows a trend of higher dT at lower θ that is only formulated from mean, seasonal conditions with actual negligible changes to dT during drying. To avoid this scenario and explicitly account for temporal evolution of each individual drydown in the phase space, two constraints on this piecewise linear model are included. (1) The median of $\Delta dT/\Delta \theta$ increments for $\theta < \theta^*$ must be negative. (2) The median of $\Delta dT/\Delta \theta$ increments for $\theta < \theta^*$ must be less than the median of $\Delta dT/\Delta \theta$ increments for $\theta > \theta^*$. Note that since we are evaluating drying periods, $\Delta \theta$ is always negative and incorporates time with consecutive decreases. These conditions state that (1) dT must generally increase in time during drying in the water-limited regime and (2) that these dT increases are typically greater in the water-limited than energy-limited regime, respectively. Effectively, θ^* is constrained to θ values that meet these conditions. If these two $\Delta dT/\Delta \theta$ increment conditions are not met for the transitional hydrologic regime, the two simpler models (water or energy limitation only) are considered. These criteria removed 22% of θ^* estimates where the simpler models were subsequently considered instead. Further, Condition (1) is also imposed for the water-limited only regime where the median of all $\Delta dT/\Delta \theta$ increments must be negative. If this is not met, only the energy-limited regime is considered. Without imposing ancillary data, these constraints set a definition for θ^* based on what is broadly expected of dT behavior during drying in water and energy-limited regimes. This ultimately ensures that when θ^* is detected, the short-term temporal evolution of dT during drydowns and not seasonal relationships formulates the relationship shown in Figure 3a.

Model fitting is validated with a 10-fold cross-validation procedure (James et al., 2014) as used in previous hydrologic regime classification studies (Akbar et al., 2018; Schwingshackl et al., 2017). This provides both a test of robustness for the AIC method and bootstrapped confidence intervals on m and θ^* . The approach is as follows. At each location, all observed (θ and dT) pairs are randomly divided into 10 equal-sized groups. Next, each model is fit to data using 9 of the 10 bins as training data to estimate their respective model parameters. Then, the fit models and parameters are compared against the data in the 10th group via a mean squared error (MSE) for each model. Analogously to AIC, this cross-validation procedure tests whether

additional model parameters result in overfitting of the data which would manifest in high MSE in comparing model to validation data in the 10th group. Each group is used once as the validation data resulting in 10 iterations. Thirty Monte Carlo replicates of this process are obtained by randomly binning the (θ and dT) pairs into the 10 groups for each of the 30 iterations. Therefore, for each of the three models, an average of 300 MSE's is computed. The "best" model is that with the lowest average MSE. Note that this procedure was only used as a validation technique.

2.4.3. Potential Confounding Variable Impacts on θ^* and m

We finally assess how potential confounding factors can influence θ^* and *m* estimates (see section 1.3 discussion). We use two independent approaches to detect shifts in these metrics: (1) multiple regression and (2) conditioning techniques.

The multiple regression technique includes fitting a multiple linear regression model to dT during drydowns:

$$dT = \beta_0 + \beta_\theta \theta + \beta_{VPD} VPD + \beta_{Wind} Wind + \beta_{Rs} Rs + \beta_\alpha \alpha + \epsilon, \tag{4}$$

where β_0 is the intercept, all other β are regression coefficients of respective dependent variable, and ε is residual error. Then, the variations of the potential confounding factors are subtracted from dT to obtain a dT time series that does not include the time variations of these factors (dT^C):

$$dT^{C} = dT - \beta_{VPD} VPD - \beta_{Wind} Wind - \beta_{Rs} Rs - \beta_{\alpha} \alpha.$$
(5)

Using the estimation procedure in 2.4.2., θ^* and *m* are estimated from both (θ and dT) and (θ and dT^C) pairs and are compared. An example is shown in Figure 3b. dT^C thus represents dT during drying periods with variability and coupling due to other factors removed. Only effects of θ principally remain. We remove all factors' variations together as in equation (5) as well as remove each individual factor's variations, where equation (5) is modified to subtract out only one factor's variations. Caveats exist in this framework including assumption of linear interactions, high covariation between regressors, and inability to account for autoregressive terms due to irreconcilable time gaps between drydowns (Gu et al., 2006). Finally, while we remove seasonality of the temporal evolution during each drydown, some seasonality may still exist within the drydown initial conditions. We repeat the analysis on the anomalies of each variable (not shown), though the interpretability of surface energy partitioning relationship with θ becomes nebulous. Overall, barring some caveats, this regression approach is attractive because it allows quantification of changes in θ^* and *m* due to all variables together and each variable individually.

We also employ a conditioning technique that is independent of regression technique assumptions. In this conditioning approach, θ^* and *m* are estimated from dT– θ relationships given only drydowns with a factor (1) significantly changing or (2) not changing. Specifically, in a given pixel, all drydowns are ranked from highest to lowest median change of a selected single factor (e.g., VPD). In the first case, θ^* and *m* are estimated for (θ and dT) pairs during drydowns with above the 60th percentile of changes in the factor (increases in variable). For the second case, θ^* and *m* are estimated for (θ and dT) pairs during drydowns (inactivity/decreases in variable). An example is shown for VPD in Figure 3c. Like in the multiple regression technique, the θ^* and *m* from both methods are compared. This evaluates how drying periods with the presence of large increases in other factors (with significant magnitude of change as will be shown later in Figure 6) propagate into amplified/subdued water-limited couplings with dT (impacting *m*) and shifting a mean state θ^* . This approach has its own shortcomings including foremost a reduced sample size from conditioning only on 40% of the data for the θ^* and *m* estimates. It also only includes conditioning on single factors at a time ignoring interactions with other variables.

Ultimately, while both methods require caveats, they provide independent, data-driven tests of how systematic variations in other factors influence θ^* and *m* estimates. Note that magnitudes of changes in θ^* or *m* are not expected to be equivalent between methods. Also, some nonlocal factors (i.e., mesoscale advection of warm air) could influence dT along with the local factors considered here, but their impacts would require use of a model-based assessment which we do not attempt in this study.





Figure 4. Region partitioning into common climates and vegetation types based on IGBP land cover classification (>75% pixel homogeneity). Note that grasslands and mosaic land cover types are combined in the Sahel region.

3. Results

3.1. Energetic Partitioning-Soil Moisture Relationship Observations

We discretize ecohydroclimatic regions by including pixels with the same International Geosphere-Biosphere Programme land cover classification (>75% homogeneity within each pixel) and collocation within similar climate regime as shown in Figure 4. Thus, each region roughly has the same vegetation type and climatic conditions. dT behavior during all drydowns within each regime is plotted together in Figure 5. General behavior is as expected from Figures 1 and 2, with dT negatively correlated with EF from previous studies (Dirmeyer et al., 2000; Koster et al., 2009; Short Gianotti et al., 2019). Biomes with predominantly herbaceous vegetation (e.g., grasslands, savannas, and shrublands) show stronger dT- θ coupling at lower soil moisture values than those with woody vegetation (e.g., woody savannas and broadleaf forests). The available energy is increasingly partitioned towards sensible heat flux with decreasing θ during the Stage II or water-limited evaporation regime. Slopes tend to be slightly lower at higher initial θ presumably due to initial well-watered conditions where most energy is partitioned into latent energy which keeps the surface temperature steadier throughout the day. Results from land surface schemes seem to generally reflect sensitivity of energy partitioning to θ at a large range of θ values (e.g., higher θ^*) for drier regions as shown here (Dirmeyer et al., 2000). Further, increased surface energy partitioning sensitivity to θ (e.g., more negative slopes) as shown in these drier regions is currently captured in reanalysis (Schwingshackl et al., 2017).

Interestingly, evergreen broadleaf forests (Congo Basin) show no dT sensitivity to θ even at low θ , despite some land surface models parameterizing a high sensitivity and low θ^* (Dirmeyer et al., 2000). Further



Figure 5. Diurnal temperature amplitude (dT; as proxy for surface energy partitioning)–soil moisture relationships during drydowns aggregated into ecohydroclimatic regions. Bold lines indicate the binned median in equal bins of $0.05 \text{-m}^3/\text{m}^3$ increments while shading bounds the 25th and 75th percentile of the empirical distribution of dT within the soil moisture bin. Progression in time of drying periods is from high to low soil moisture.



Figure 6. Time evolution of other factors during soil moisture drydowns. (a) Observed downward solar radiation flux. (b) Observed surface albedo. (c) Reanalysis wind speed. (d) Reanalysis vapor pressure deficit at 1:30 PM. Progression in time of drying periods is from high to low soil moisture.

investigations reveal that low θ satellite observations in the Congo Basin are likely the result of noisy retrievals. The behavior in evergreen broadleaf forests should therefore be treated with caution as low θ (water limitation) may not have been actually observed in this region during the study period.

The behavior in Figure 5 is not due to seasonal mean states. Evaluating individual drydowns on this phase space shows that the $dT-\theta$ relationships are formulated from the time evolution of dT increases during the 10–20 days of drying. Note that progression in time is from high to low θ . Again, seasonal trends are removed from these relationships indicating that the behavior is not due to phase differences associated with seasonal cycles (Williams & Torn, 2015).

3.2. Potential Effects of Confounding Factors

Directly addressing how other relevant factors evolve following a rain event (see section 1.3 and Figure 1), Figure 6 shows short-term behavior of Rs, α , wind speed, and VPD during drying with progression in time from high to low θ . Observed Rs in Figure 6a shows large increases during drying following rain events. Tests with SEVIRI cloud reflectance and MERRA-2 reanalysis cloud cover information revealed that this is due to progressively decreasing cloud cover after precipitation ceases. SMAP's low spatial resolution is likely sampling mesoscale events rather than smaller scale, convective cells. The temporal persistence of cloud cover is generally greater for these mesoscale events as observed in the Sahel (Mathon & Laurent, 2001). Some studies noted the effects of solar radiation on temperature (Teuling et al., 2006), while others attempt to remove its effects (Miralles et al., 2012). Regardless, one can hypothesize that the Rs increases could amplify partitioning into sensible heating (and ground heat flux) during drying. Note, the Rs decrease at high initial θ in open shrublands is likely an artifact of averaging on the phase space as all individual Rs trajectories increase under all θ (not shown).

Observed α in Figure 6b, at least in the driest biomes, appears to show a ~0.03 increase during drying periods consistent with previous in situ investigations in the US Great Plains (Amano & Salvucci, 1999). α is expected





Figure 7. (a) θ^* estimation from dT observations (m³ m⁻³). (b) *m*, or dT change [K] per θ decrease of 0.1 m³/m³ during a drydown (dT increases during drying are positive). Both figures include the spatial distribution of values in the inset. (c) Fraction of time spent in water-limited regime. Hydrologic regime classifications based on observed dT– θ relationships are shown in the inset. Stippling indicates water-limited regime where *m* could not be estimated due to low variations in θ ($\sigma(\theta) < 0.03 \text{ m}^3/\text{m}^3$).

to increase from drying soil and desiccating vegetation. Wind speed exhibits a tendency to increase during drying under drier surfaces in Figure 6c. Bare soil modeling results showed that increasing wind speed could strongly shift θ^* to wetter values (Haghighi et al., 2018). Finally, VPD increases significantly due to both increasing temperature and decreasing humidity as shown in Figure 6d. These effects are likely coupled to amplifying sensible heating (as well as ground heat flux) during surface drying. Overall, the Figure 6 behavior indicates that factors other than θ may systematically explain the dT– θ trends and scatter shown in Figure 5.

3.3. θ^* , *m*, and Hydrologic Regime Estimation

Estimated θ^* from observations (θ and dT) are shown in Figure 7a from those pixels robustly identified as having transitional regimes in Figure 3a. Areas without an identifiable θ^* are characterized by either a persistent water-limited or energy-limited evaporation regimes. Spatial patterns of θ^* are coherent and show a tendency to decrease with sand fraction ($\rho = -0.43$; p < 0.01) similarly to Akbar et al. (2018). While the time it takes to reach θ^* may be quicker in coarse soils during drying (Aminzadeh & Or, 2014), moisture extraction via ET can occur at lower θ due to less matric suction than in finer soils. This is consistent with findings of lower θ thresholds on plant water uptake for coarser soils (Feldman et al., 2018). θ^* and mean annual precipitation are weakly, positively correlated ($\rho = 0.26$; p < 0.01) with the sign of dependence consistent with previous studies (Akbar et al., 2018; Schwingshackl et al., 2017). Reanalysis and model estimates show a much stronger climate dependence with higher θ^* generally closer to the equator, especially in Africa (Schwingshackl et al., 2017). In contrast, these estimates here interestingly indicate that all biomes have similar θ^* distributions (see Table 1), despite a wide range of mean annual precipitation values. Pairwise Kolmogorov–Smirnov tests indicate that grasslands and woody savannas have θ^* spatial distributions typically significantly different from the other biomes' distributions (p < 0.05) (Massey, 1951). However, the



Land Cover Dependence of Surface Energy Partitioning Characteristics and θ Retrieval Uncertainty

	Shrublands	Grasslands	Savanna	Woody Savanna	EBF
$\theta^* [\mathrm{m}^3 \mathrm{m}^{-3}]$	0.14 (0.06)	0.12 (0.05)	0.15 (0.06)	0.14 (0.07)	0.18 (0.05)
ΔdT per θ decrease (m) [K]	8.9 (0.9)	9.4 (0.6)	6.0(0.8)	4.7 (0.6)	1.0(0.1)
Time Spent in Stage II ET [%]	94% (6%)	84% (11%)	100% (30%)	66% (33%)	0% (26%)
Mean Annual Precipitation [mm]	267	584	625	1265	1736
$E[TB_H - TB_V][K]$	20.8	24.6	15.2	11.3	3.1
σ (TB _H) [K]	11.7	17.3	7.4	6.8	2.4
σ (TB _V) [K]	8.8	12.4	5.8	4.6	2.0

Note. All values are reported in spatial medians with the spatial standard deviation in parenthesis. *m* is reported as dT change [K] per θ decrease of 0.1 m³/m³ during a drydown (dT increases during drying are positive). The lower rows display brightness temperature (TB) mean difference between polarization and standard deviations. Larger values of TB metrics indicate greater θ retrieval robustness. EBF stands for evergreen broadleaf forest and TB for brightness temperature.

 θ^* medians and standard deviations of each biome (as well as their probability density functions; not shown) are all broadly similar (see Table 1). Therefore, θ^* itself is somewhat independent of the mean moisture state. To a first order, one would expect lower θ^* in humid regions that have generally greater vegetation and less bare soil cover. Root systems are typically accessing deeper soil water pools than the soil surface interface (as represented by SMAP θ) where bare soil evaporation occurs. These deeper water pools have slower changing dynamics and may allow transpiration to continue at its potential even after evaporation from the soil surface interface becomes water limited. However, a study contrasting semiarid savanna and grassland appear to show θ^* of near 0.15 m³/m³ for both biomes, supporting some independence of vegetation cover on θ^* (Baldocchi et al., 2004). Thus, while a spatially coherent θ^* pattern (evidence for some spatial autocorrelation in Figure 7a instead of complete randomness) emerges here, coarse climate considerations do not explain it.

The slope, *m*, is greatest in transitional regions between humid and arid climates as shown in Figure 7b. This appears to be captured in some reanalysis outputs (Schwingshackl et al., 2017) but is inconsistent with some land surface schemes that are parameterized with greater m in wetter, forested regions than for drier, less vegetated regions (Dirmeyer et al., 2000). Generally, higher m is expected in drier regions with herbaceous vegetation cover because of (1) vegetation with more responsiveness to individual rainfall events (and less stomatal control) and (2) lower vegetation cover. In tropical dry regions, vegetation hydraulic systems largely respond to individual rainfall events (Feldman et al., 2018). Thus, a transpiration response is expected given a perturbation of θ . In contrast, woody vegetation in more humid environments may be exhibiting greater stomatal control under water limitation (Konings & Gentine, 2017; Lin et al., 2015), accessing deeper water stores not observed by the satellite and/or showing generally less responsiveness to changes in surface soil water availability. These scenarios could reduce m. Also, the drier regions generally have a significant contribution from bare soil due to patchy or light vegetation cover. Bare soil is always responding to changes in θ under water limitation. These factors could in part explain the spatial pattern in Figure 7b. Note that for simplicity, *m* is reported as the increase in dT given a unit soil moisture decrease of $0.1 \text{ m}^3/\text{m}^3$ during a drydown. The patterns of *m* are interpretable as a proxy for coupling between the surface water and energy balances without impositions of land-atmosphere model assumptions. Given physical differences between dT and EF, further work is required to evaluate whether m magnitudes uniquely relate to $\frac{\partial EF}{2\alpha}$ magnitudes.

The 10-fold cross-validation procedure indicates that the θ^* and *m* estimates with AIC are robust (Figures S2 and S3). The spatial median θ^* bootstrapped standard deviation is 0.004 m³/m³. Additionally, AIC estimated θ^* is within 0.01 m³/m³ of the mean θ^* from the 10-fold cross validation replicates 86% of the time. Similarly, the median *m* standard deviation is 0.3 K. AIC estimated *m* is within 1 K of the mean *m* from the 10-fold cross validation replicates 83% of the time. A slight bias between AIC and cross-validation *m* estimation exists near the Kalahari Desert in Southern Africa. The 10-fold cross-validation procedure selected the transitional model regime 59% of the time, while the AIC method selected it 57% of the time. Thus, AIC is at least marginally conservative in selecting transitional regimes, the most complex model. Therefore, the AIC approach is deemed sufficient for use in this study, especially for the purposes of evaluating θ^* and *m* parameters rather than details of the hydrologic regime classification. This justifies the sole use of the AIC



Figure 8. Effects of other factors on θ^* . (a) Difference in θ^* including minus removing effects of other factors (θ^* estimated from dT minus θ^* estimated from dT^C equation (5)) [m³ m⁻³]. (b) Relative contribution of each factor to the change in θ^* . Note that wind speed is the lowest contributor to θ^* changes and is thus not included in this plot. (c–g) Spatial distribution of change in θ^* including minus removing effects of all cofactors, only VPD, only α , only Rs, and only wind, respectively. In other words, all differences can be interpreted as direction of change in θ^* due to variations in the respective variables. Rs= insolation; VPD=vapor pressure deficit; ∂ =surface albedo; θ =soil moisture.

approach in quantifying effects of other factors on θ^* and *m*. Model RSS from AIC are also relatively uniform throughout the study region (see Figure S4).

One utility of observation-based θ^* estimates is to assess where the local environment varies across this threshold and transitions from uncoupled to coupled water and energy balance regimes. We estimate the fraction of time in the water-limited regime or percentage of time the observed θ is below θ^* (see Figure 7 c). The dominant regime classification is shown in the inset of this figure. The dominant regimes observed broadly corroborate previous findings from land surface models (Koster et al., 2009; Schwingshackl et al., 2017). However, regime dominance only reflects the spatiotemporal scales of the datasets. Any given location from rainforest to desert will experience energy and water-limited regimes at some point given sufficient years of data at a frequent sampling interval (i.e., hourly). Instead, the time spent in a given regime holds more information as a continuous metric with less dependence on the dataset's spatiotemporal scales (scales may still impact this metric's robustness). Note that this uses the full θ time series to avoid biases from inconsistent drydown sampling between pixels. A spatial gradient is observed with drier regions generally spending more time in the water-limited regime as expected (Table 1). It is worth noting that biomes in the Sahel tend to exhibit collocation of frequent transitioning between hydrologic regimes and large m (compare Figures 7b and 7c). The Sahelian grasslands spend on average 84% of the time in the water-limited regime. However, when considering only drydowns that occur primarily in the wet season, 66% of the time is spent in the water-limited regime reflecting rainfall regimes that promote frequent transitions between Stage I and II ET. This indicates that ET may be often close to potential ET following rainfall when in the water-limited

regime. This suggests a potential for land surface control on the lower atmosphere as there is both strong energetic partitioning sensitivity to θ and likely sufficiently high ET (Koster et al., 2004). Many previous studies labeled this region as a hotspot showing strong soil moisture–climate coupling (Dirmeyer, 2011; Shinoda & Yamaguchi, 2003) and probable soil moisture-precipitation feedbacks (Koster et al., 2004). In contrast, woody savannas south of the Congo Basin also exhibit frequent transitioning between energy and water limitation, but *m* is lower on average. Spatial variability of both metrics is greater south of the equator with potentially isolated coupling hotspots.

 θ retrieval uncertainty is higher in dense broadleaf forests and woody savannas as shown in Table 1. Here, lower mean difference and standard deviation of each brightness temperature (TB) polarization (wave orientations used to estimate θ) indicate lower signal to noise ratio. Both of these metrics need to be sufficiently high to robustly retrieve θ . Specifically, high mean difference between TB polarizations allows partitioning between soil and vegetation surface emission, and high TB standard deviation (larger dynamic range) results in less noise in θ estimation. Land surface temperatures used to derive dT from SEVIRI have biases of less than 1 K throughout vegetated biomes (Freitas et al., 2010; Göttsche et al., 2016; Trigo et al., 2008). Instead, variations in humidity and aerosols optically thicken the atmosphere and result in significant uncertainty (Freitas et al., 2010; Göttsche et al., 2016). Also, the retrieved temperature characterizes the surface skin which may represent the canopy crown in forests as opposed to the soil surface. Soil-vegetation temperature disequilibrium can indeed occur during the day. Ultimately, broadleaf forest and less so woody savanna results should be considered with caution.

3.4. Effects of Other Factors on θ^* and *m* Estimates

Regression approach results are discussed first, quantifying the relative effects of VPD, α , Rs, and wind speed (both individually and altogether) on *m* and θ^* . The conditioning approach is used to corroborate individual factor effects on *m* and θ^* as determined in the regression approach. The metrics are interpreted as the shift in *m* and θ^* due to VPD, wind speed, α , and/or Rs variations.

As shown in Figure 8a, the variations of all factors together as determined from the regression approach tended to reduce θ^* in general with a median 0.01 m³/m³ decrease. VPD variations have the greatest influence on this θ^* decrease as shown in Figures 8b and 8d (see also Table S1). This VPD effect occurred in semiarid regions such as the Sahel. Elsewhere, the relative effects of factors are random showing that wind speed, α , and Rs show no consistent, significant impact on θ^* .

Analogously, Figure 9 shows these factors' influence on *m*. Their variations tend to increase dT sensitivity to θ by about 2 K per 0.1 m³/m³ of soil drying compared to not including their variations. In this case, the effects of all variables on *m* show more coherent spatial patterns than for θ^* . VPD again dominates contributions to the increase in *m* in subhumid regions. However, solar radiation and α do show coherent, dominant impacts in the arid regions in Southern Africa and boundaries of the Sahara Desert.

To corroborate the regression approach findings of shifts in θ^* and *m* due to individual factors, we compare to conditioning approach results. We evaluate whether an individual factor's variations changed θ^* or m(1)significantly and (2) in the same direction for both independent methods (regression and conditioning approaches). This is evaluated for all pixels in Africa as well as pixels in each ecohydroclimatic region. Evergreen broadleaf forests are excluded due to predominant detection of energy limitation. For each biome and potential confounding factor pair, a Mann-Whitney U test (Mann & Whitney, 1947) is performed between spatial θ^* distributions with the factor's variations included and removed. This is repeated for m spatial distributions. This nonparametric test is performed as Jarque-Bera normality tests (Jarque & Bera, 1980) reveal all θ^* and m spatial distributions are non-Gaussian. Results are shown in Table S1 and summarized here. Both regression and conditioning approaches independently show that θ^* decreased (~0.01 m³/ m³) in the Sahelian grasslands when VPD increased during drydowns. In other words, increasing atmospheric demand during the drydown shifts the water-limited regime transition to a lower θ . However, a robust shift in θ^* is not detected in other biomes due to any other factor. *m* increases due to VPD are on the order of 1 K and are more widespread across generally drier biomes. Rs increases also are associated with the same magnitude of increase m but only in shrublands of Southern Africa. No consistent, significant effects of α and wind speed are found. This is likely due to minimal α dynamics and inconsistent wind speed influences on energy partitioning during drying.



Figure 9. Same as Figure 8 but for *m*, or dT change [K] per θ decrease of 0.1 m³/m³, in water-limited regime. Rs= insolation; VPD=vapor pressure deficit.

These results suggest that (1) there is a spatial dependence on where these factors tend to impact surface energy partitioning as coupled to θ on weather timescales. (2) VPD and Rs are the only variables that are robustly identified as confounding the surface energy partitioning- θ relationship. (3) The actual impacts of these factors on θ^* and m are typically minor. Ultimately, despite clear variations in these variables during transitions from energy to water limitation as in Figure 6, moisture availability remains to be the dominant factor in partitioning surface energy. Specifically, VPD appears to influence θ^* and *m* more than Rs, α , and wind speed, but these VPD variations are highly coupled to θ . VPD, the difference between saturation vapor pressure and actual vapor pressure, increases with warmer and drier air conditions. Soil moisture contributes to both of these factors: decreasing latent heat flux favors sensible and ground heat flux for dissipating available energy. This warms the overlying air and soil and reduces ET, decreasing humidity. Considering the multiple regression coefficients multiplied by their respective factor's standard deviation, VPD and θ variations are coupled to the largest changes in dT (~2 K) with typically little influence from the other factors (~0.1 K). External forcings, such as Rs and wind speed, therefore do not appear to have widespread, consistent influences on surface energy partitioning, despite some coupling to θ as shown in Figure 6. Further work is required to separate effects of surface and atmospheric moisture availability and their relative contributions to surface energy partitioning. We caution that these results do not suggest that time evolution of external forcings can be ignored in coupling analyses, for example. As seen in Figures 8 and 9, there are cases where confounding effects on surface energy partitioning are potentially large with θ^* and *m* changing in time at a given location. Further work should also investigate whether these results are sensitive to spatiotemporal scales. It is suspected that anomalous forcings become more apparent on energetic partitioning– θ relationships at finer spatiotemporal scales. Indeed, pore-scale bare soil modeling studies showed vast variations in θ^* especially with temperature and wind speed (Haghighi et al., 2018).

It is interesting that VPD anomalous increases tend to decrease θ^* . One might expect the opposite where water limitation would be reached at a wetter condition during drying. Assuming similar relationships between θ^* and potential confounding variables in both ET and dT space, θ^* therefore relies on the interplay between actual and potential ET. Specifically, θ^* should be reached during drying when actual ET initially decreases below potential ET. All potential confounding factors generally increase during drying (see Figure 6), and all should contribute to an increasing potential ET. Thus, if actual ET hypothetically remained constant or decreased while potential ET increased during drying, θ^* would increase. However, a decrease in θ^* suggests that VPD anomalously increasing is due to actual ET continuing near its potential under drier surface conditions. In other words, VPD increasing might cause both actual and potential ET to increase together during drying such that θ^* decreases.

For *m*, we postulate that the increases in *m* due to anomalous increases in VPD and Rs are likely due to amplified sensible heating. In comparing Figures 5 and 6, increasing VPD and insolation both increase along with dT under water limitation which would suggest that they could amplify dT increases during drying.

It is important to note that this analysis is conducted on the time evolution of dT changes with drying soil. This analysis (Figures 8 and 9 and Tables S1) quantifies how short-term (1-3 day) changes, rather than mean state, of other factors influence dT– θ relationships. Conditioning dT– θ relationships on the mean state of net radiation (high or low), for example, would only suggest how dT– θ relationships are different between seasons. Since net radiation is coupled to nearly every other factor between land surface and boundary layer (Betts, 2004), this would say little about net radiation's direct influence on surface energy partitioning. Thus, conducting the analysis on the variables' short-term time evolution should be relatively isolated from mean climate spurious relationships.

4. Discussion

With VPD and Rs increases during drying on subweekly scales as in Figure 6, it is plausible that positive feedbacks between these factors can be detrimental to vegetation health. At low θ , vegetation water content at large scales appears to decrease following rainfall along with θ (Feldman et al., 2018). Thus, the sequence of increasing incoming shortwave radiation, increasing atmospheric demand, increasing sensible heating, and/or drying soil may exacerbate vegetation function over days to weeks given insufficient rainfall. This has implications for θ limited transpiration which also has a positively feedback that further limits rainfall (Green et al., 2017). There is thus cause to evaluate vegetation parameter time evolution within this multivariable framework in addition to its influence on land-atmosphere coupling (Zscheischler et al., 2015).

With advances in raw and assimilated remote sensing soil moisture, soil moisture–temperature coupling estimates are becoming more robust (Dong & Crow, 2019, 2018), replacing previously used precipitation-temperature metrics. The dT– θ relationships found here present caution to soil moisture–temperature coupling metrics. Above θ^* , θ controls on afternoon surface temperature, like dT, are subdued (not shown). Thus, coupling metrics may need to carefully account for differences in θ controls on temperature considering water and energy-limited regimes instead of all regimes together.

The "triangle method" has been used to estimate θ from satellite optical vegetation information and surface temperature (Carlson et al., 1994). The changing sensitivity of dT to θ in in the water-limited regime due to changes in insolation and VPD can confound this θ estimation technique.

Implications for land-atmosphere feedbacks are only for the terrestrial segment, with soil moisture influencing the surface latent and sensible heat fluxes (Dirmeyer, 2011). The time evolution during 10–20-day drying periods following rainfall shows a widespread tightly coupled system of soil water loss and increasing atmospheric demand, sensible heat, and incoming radiation. Despite this coupling, θ^* and *m* appear to sustain only minor systematic shifts at spatiotemporal scales of land surface models. Further, under coarse scale considerations, it is noteworthy that the θ^* distribution is relatively narrow with similar values in each ecohydroclimatic regime. Therefore, moderate moisture regimes varying just under θ^* with high *m* are expected to show strong land influence on the lower atmosphere. This appears to occur broadly across the Sahel region. In contrast, minimal land control on the lower atmosphere is expected in humid regions that have θ variability above θ^* (e.g., energy limited).

5. Conclusions

We use two remote sensing datasets to examine key aspects of the relationship between surface water and energy balances over land surfaces. We use satellite-based dT range from a sensor in geostationary orbit and soil moisture (θ) from a low-frequency sensor in low-Earth orbit with shared coverage of diverse climates and biomes across the African continent. These two datasets are independent of any models or parameterizations linking the water and energy balances. They are used to quantify key characteristics of this coupling using the time evolution of dT and θ observations during drying periods: (1) the soil moisture threshold (θ^*) that marks the transition from energy-limited to water-limited evaporation regimes and (2) the strength of the coupling between dT and θ in the water-limited regime. On average, ecohydroclimatic regions with drier conditions and herbaceous vegetation cover tend to exhibit stronger coupling of water and energy balances during water-limited ET regimes, while the θ^* values and patterns are similar under nearly all of the diverse ecohydroclimate regions of Africa. The regions with the highest land surface flux partitioning sensitivity to θ are interestingly collocated with regions that frequently transition between energy and water limitation. This specifically occurs in the Sahelian grasslands, a region previously receiving much attention as a land-atmosphere feedback hotspot.

The results shown here are purposely free of land surface model and parameterization influences. Estimates of these key characteristics based on land surface model outputs may only reflect explicit and/or implicit model formulations. These estimates provide observations of θ control on energetic partitioning (as commonly visualized in the EF- θ relationship) at the spatial scales of land surface models, noting that dT is used here as proxy for sensible heat flux.

We also assess the possible influence of other environmental variables on dT during drydowns. Vapor pressure deficit, α , wind speed, and insolation generally significantly increase across Africa during drying periods. In order to test how these variations may confound surface energy partitioning $-\theta$ relationships, we use regression and conditioning approaches to test whether effects of each factor shift the value of our study metrics. Ultimately, only vapor pressure deficit and less so insolation tend to have any influence on these relationships. Specifically, anomalous increases in both variables tend to increase dT by approximately 1 K per 0.1 (m³ m⁻³) of soil drying. Vapor pressure deficit increases tend to reduce θ^* by 0.01 (m³ m⁻³) though only in Sahelian grasslands. Vapor pressure deficit and insolation thus show greater potential as confounding factors than α and wind speed. The median 0.01 m³ m⁻³ and 1 K shifts in θ^* and m are, however, arguably small, though isolated cases show larger changes. Additionally, vapor pressure deficit is strongly coupled to θ which confounds effects of direct control of humidity deficits on surface energy partitioning as separated from soil water deficits. Ultimately, both surface and atmospheric moisture availability together strongly constrain surface energy partitioning despite significant time evolution of other surface and atmospheric variables during these drying periods. We suspect some dependence of results on spatiotemporal scales. More advanced sensors with finer spatiotemporal resolutions will likely be able to detect stronger, systematic effects of confounding factors on energetic partitioning- θ relationships.

Acknowledgments

The authors acknowledge funding from NASA in the form of a sponsored research grant (Subcontract No. 1510842), SMAP L1C brightness temperature used to retrieve soil moisture are available from the National Snow and Ice Data Center (NSIDC) (https://nsidc.org/data/ SPL1CTB E). All LandSAF datasets are available from EUMETSAT (https:// landsaf.ipma.pt). MERRA-2 reanalysis datasets are available from NASA's Global Modeling and Assimilation Office (https://gmao.gsfc.nasa.gov/ reanalysis/MERRA-2/data_access/). Generated matfiles plotted in this study are available at https://github.com/ afeld24/dTSoilMoistureRelations. This work used eddy covariance data acquired and shared by the FLUXNET community, including the ZA-Kru Skukuza, South Africa site (DOI: 10.18140/FLX/1440188). The authors thank two anonymous reviewers for their keen, constructive comments.

References

Akaike, H. (1974). A new look at the statistical model identification. *IEEE Transactions on Automatic Control*, 19(6), 716–723. https://doi.org/10.1109/TAC.1974.1100705

Akbar, R., Short Gianotti, D. J., McColl, K. A., Haghighi, E., Salvucci, G. D., & Entekhabi, D. (2018). Estimation of landscape soil water losses from satellite observations of soil moisture. *Journal of Hydrometeorology*, 19, 871–889. https://doi.org/10.1175/JHM-D-17-0200.1

Amano, E., & Salvucci, G. D. (1999). Detection and use of three signatures of soil-limited evaporation. *Remote Sensing of Environment*, 67, 108–122. https://doi.org/10.1016/S0034-4257(98)00076-5

Aminzadeh, M., & Or, D. (2014). Energy partitioning dynamics of drying terrestrial surfaces. Journal of Hydrology, 519, 1257–1270. https:// doi.org/10.1016/j.jhydrol.2014.08.037

Baldocchi, D. D., Xu, L., & Kiang, N. (2004). How plant functional-type, weather, seasonal drought, and soil physical properties alter water and energy fluxes of an oak-grass savanna and an annual grassland. Agricultural and Forest Meteorology, 123, 13–39. https://doi.org/ 10.1016/j.agrformet.2003.11.006

Bateni, S. M., & Entekhabi, D. (2012). Relative efficiency of land surface energy balance components. *Water Resources Research*, 48, W04510. https://doi.org/10.1029/2011WR011357

Betts, A. K. (2004). Understanding hydrometeorology using global models. Bulletin of the American Meteorological Society, 85(11), 1673–1688. https://doi.org/10.1175/BAMS-85-11-1673

Betts, A. K., Ball, J. H., Beljaars, A. C. M., Miller, M. J., & Viterbo, P. A. (1996). The land surface-atmosphere interaction: A review based on observational and global modeling perspectives. *Journal of Geophysical Research*, 101(D3), 7209–7225. https://doi.org/10.1029/ 95JD02135

- Betts, A. K., Desjardins, R., Worth, D., & Beckage, B. (2014). Climate coupling between temperature, humidity, precipitation, and cloud cover over the Canadian Prairies. *Journal of Geophysical Research: Atmospheres*, 119, 13,305–13,326. https://doi.org/10.1002/ 2014JD022511.Received
- Betts, A. K., Desjardins, R., Worth, D., & Cerkowniak, D. (2013). Impact of land use change on the diurnal cycle climate of the Canadian Prairies. Journal of Geophysical Research: Atmospheres, 118, 11,996–12,011. https://doi.org/10.1002/2013JD020717
- Brubaker, K. L., & Entekhabi, D. (1996). Analysis of feedback mechanisms in land-atmosphere interaction. *Water Resources Research, 32*, 1343–1357. https://doi.org/10.1029/96WR00005
- Budyko, M. I. (1974). Climate and Life. New York: Academic Press.
- Cao, B., Liu, Q., Du, Y., Roujean, J.-L., Gastellu-Etchegorry, J.-P., Trigo, I. F., et al. (2019). A review of earth surface thermal radiation directionality observing and modeling: Historical development, current status and perspectives. *Remote Sensing of Environment, 232*, 111304. https://doi.org/10.1016/j.rse.2019.111304
- Carlson, T. N., Gillies, R. R., & Perry, E. M. (1994). A method to make use of thermal infrared temperature and NDVI measurements to infer surface soil water content and fractional vegetation cover. *Remote Sensing Reviews*, 9, 161–173. https://doi.org/10.1080/ 02757259409532220
- Carrer, D., Lafont, S., Roujean, J.-L., Calvet, J.-C., Meurey, C., Le Moigne, P., & Trigo, I. F. (2011). Incoming solar and infrared radiation derived from METEOSAT: Impact on the modeled land water and energy budget over France. *Journal of Hydrometeorology*, 13, 504–520. https://doi.org/10.1175/jhm-d-11-059.1
- Carrer, D., Moparthy, S., Lellouch, G., Ceamanos, X., Pinault, F., Freitas, S. C., & Trigo, I. F. (2018). Land surface albedo derived on a ten daily basis from Meteosat Second Generation Observations: The NRT and climate data record collections from the EUMETSAT LSA SAF. Remote Sensing, 10(8), 1262. https://doi.org/10.3390/rs10081262
- Castelli, F., Entekhabi, D., & Caporali, E. (1999). Estimation of surface heat flux and an index of soil moisture using adjoint-state surface energy balance. *Water Resources Research*, *35*, 3115–3125. https://doi.org/10.1029/1999WR900140
- Ceamanos, X., Carrer, D., & Roujean, J. L. (2014). Improved retrieval of direct and diffuse downwelling surface shortwave flux in cloudless atmosphere using dynamic estimates of aerosol content and type: Application to the LSA-SAF project. Atmospheric Chemistry and Physics, 14, 8209–8232. https://doi.org/10.5194/acp-14-8209-2014
- Chan, S. K., Bindlish, R., O'Neill, P. E., Njoku, E., Jackson, T., Colliander, A., et al. (2016). Assessment of the SMAP passive soil moisture product. *IEEE Transactions on Geoscience and Remote Sensing*, 54, 4994–5007. https://doi.org/10.1109/TGRS.2016.2561938
- Charney, J. G. (1975). Quarterly meteorological. Quarterly Journal of the Royal Meteorological Society, 101(428), 193–202. https://doi.org/ 10.1002/qj.49710142802
- Chaubell, J., Chan, S., Dunbar, R.S., Peng, J., Yueh, S., 2016. SMAP enhanced L1C radiometer half-orbit 9 km EASE-grid brightness temperatures, Version 1. Boulder, Colorado USA. NASA National Snow and Ice Data Center Distributed Active Archive Center. doi: https://doi.org/10.5067/2C909KT6JAWS. 08.30.2017.

Chaubell, J., Yueh, S., Entekhabi, D., Peng, J., 2016. Resolution enhancement of SMAP radiometer data using the Backus Gilbert optimum interpolation technique. Int. Geosci. Remote Sens. Symp. 2016-Novem. 284–287. https://doi.org/10.1109/IGARSS.2016.7729065

- Crow, W. T., & Wood, E. F. (2002). Impact of soil moisture aggregation on surface energy flux prediction during SGP' 97. Geophysical Research Letters, 29(1), 1008. https://doi.org/10.1029/2001GL013796
- Dirmeyer, P. A. (2011). The terrestrial segment of soil moisture-climate coupling. Geophysical Research Letters, 38, L16702. https://doi.org/ 10.1029/2011GL048268
- Dirmeyer, P. A., Koster, R. D., & Guo, Z. (2007). Do global models properly represent the feedback between land and atmosphere? *Journal of Hydrometeorology*, 7, 1177–1198. https://doi.org/10.1175/jhm532.1
- Dirmeyer, P. A., Zeng, F. J., Ducharne, A., Morrill, J. C., & Koster, R. D. (2000). The sensitivity of surface fluxes to soil water content in three land surface schemes. *Journal of Hydrometeorology*, 1, 121–134. https://doi.org/10.1175/1525-7541(2000)001<0121:TSOSFT>2.0.CO;2
- Dong, J., & Crow, W. T. (2018). The added value of assimilating remotely sensed soil moisture for estimating summertime soil moisture-air temperature coupling strength. *Water Resources Research*, 54, 6072–6084. https://doi.org/10.1029/2018WR022619
- Dong, J., & Crow, W. T. (2019). L-band remote-sensing increases sampled levels of global soil moisture-air temperature coupling strength. Remote Sensing of Environment, 220, 51–58. https://doi.org/10.1016/j.rse.2018.10.024
- Eagleson, P. S. (1978). Climate, soil, and vegetation 4. The expected value of annual evapotranspiration. *Water Resources Research*, 14(5), 731–739. https://doi.org/10.1029/WR014i005p00731
- Entekhabi, D., Njoku, E. G., O'Neill, P. E., Kellogg, K. H., Crow, W. T., Edelstein, W. N., et al. (2010). The soil moisture active passive (SMAP) mission. *Proceedings of the IEEE*, 98, 704–716. https://doi.org/10.1109/JPROC.2010.2043918
- Entekhabi, D., Rodriguez-iturbe, I., & Castelli, F. (1996). Mutual interaction of soil moisture state and atmospheric processes. Journal of Hydrology, 184(1-2), 3–17. https://doi.org/10.1016/0022-1694(95)02965-6
- Ermida, S. L., Trigo, I. F., DaCamara, C. C., & Pires, A. C. (2018). A methodology to simulate LST directional effects based on parametric models and landscape properties. *Remote Sensing*, 10, 1–19. https://doi.org/10.3390/rs10071114
- Feldman, A. F., Short Gianotti, D. J., Konings, A. G., McColl, K. A., Akbar, R., Salvucci, G. D., & Entekhabi, D. (2018). Moisture pulse-

reserve in the soil-plant continuum observed across biomes. *Nature Plants*, 4(12), 1026–1033. https://doi.org/10.1038/s41477-018-0304-9 Findell, K. L., & Eltahir, E. A. B. (2003). Atmospheric controls on soil moisture–boundary layer interactions. Part I: Framework develop-

- ment. *Journal of Hydrometeorology*, *4*, 552–569. https://doi.org/10.1175/1525-7541(2003)004<0552:acosml>2.0.co;2 Findell, L., Eltahir, A. B., & Parsons, R. M. (1997). An analysis of the soil moisture-rainfall feedback, based on direct observations from
- Illinois. Water Resources Research, 33(4), 725–735. https://doi.org/10.1029/96WR03756
 Ford, T. W., Wulff, C. O., & Quiring, S. M. (2014). Assessment of observed and model-derived soil moisture-evaporative fraction relationships over the United States Southern Great Plains. Journal of Geophysical Research: Atmospheres, 119, 6279–6291. https://doi.org/ 10.1002/2013JD020225. Received
- Freitas, S. C., Trigo, I. F., Bioucas-Dias, J. M., & Göttsche, F. M. (2010). Quantifying the uncertainty of land surface temperature retrievals from SEVIRI/Meteosat. *IEEE Transactions on Geoscience and Remote Sensing*, 48, 523–534. https://doi.org/10.1109/ TGRS.2009.2027697
- Gallego-Elvira, B., Taylor, C. M., Harris, P. P., Ghent, D., Veal, K. L., & Folwell, S. S. (2016). Global observational diagnosis of soil moisture control on the land surface energy balance. *Geophysical Research Letters*, 43, 2623–2631. https://doi.org/10.1002/2014GL059364. Received
- Geiger, B., Carrer, D., Franchistéguy, L., Roujean, J. L., & Meurey, C. (2008). Land surface albedo derived on a daily basis from meteosat second generation observations. *IEEE Transactions on Geoscience and Remote Sensing*, 46, 3841–3856. https://doi.org/10.1109/ TGRS.2008.2001798

- Geiger, B., Meurey, C., Lajas, D., Franchisteguy, L., Carrer, D., & Roujean, J.-L. (2008). Near real-time provision of downwelling shortwave radiation estimates derived from satellite observations. *Meteorological Applications*, 15, 411–420. https://doi.org/10.1002/met Gentine, P., Entekhabi, D., Chehbouni, A., Boulet, G., & Duchemin, B. (2007). Analysis of evaporative fraction diurnal behaviour.
- Agricultural and Forest Meteorology, 143, 13–29. https://doi.org/10.1016/j.agrformet.2006.11.002
- Gevaert, A. I., Miralles, D. G., de Jeu, R. A. M., Schellekens, J., & Dolman, A. J. (2018). Soil moisture-temperature coupling in a set of land surface models. *Journal of Geophysical Research: Atmospheres*, 123, 1481–1498. https://doi.org/10.1002/2016JD025907
- GMAO, 2008. tavg1_2d_flx_Nx: MERRA 2D IAU Diagnostic, Surface Fluxes, Time Average 1-hourly V5.2.0, Greenbelt, MD, USA, Goddard Earth Sciences Data and Information Services Center (GES DISC), Accessed: [01.10.19].
- GMAO, 2015. MERRA-2 inst1_2d_asm_Nx: 2d,1-Hourly,Instantaneous,Single-Level,Assimilation,Single-Level Diagnostics V5.12.4, Greenbelt, MD, USA, Goddard Earth Sciences Data and Information Services Center (GES DISC), Accessed: [01.10.19], https://doi.org/ 10.5067/3Z173KIE2TPD.
- Göttsche, F. M., Olesen, F. S., Trigo, I. F., Bork-Unkelbach, A., & Martin, M. A. (2016). Long term validation of land surface temperature retrieved from MSG/SEVIRI with continuous in-situ measurements in Africa. *Remote Sensing*, 8(5), 410. https://doi.org/10.3390/ rs8050410
- Green, J. K., Konings, A. G., Alemohammad, S. H., Berry, J., Entekhabi, D., Kolassa, J., et al. (2017). Regionally strong feedbacks between the atmosphere and terrestrial biosphere. *Nature Geoscience*, *10*(Iss 6), 410–414. https://doi.org/10.1038/ngeo2957
- Gu, L., Meyers, T., Pallardy, S. G., Hanson, P. J., Yang, B., Heuer, M., et al. (2006). Direct and indirect effects of atmospheric conditions and soil moisture on surface energy partitioning revealed by a prolonged drought at a temperate forest site. *Journal of Geophysical Research*, 111, D16102. https://doi.org/10.1029/2006JD007161
- Guo, Z., Dirmeyer, P. A., Koster, R. D., Sud, Y. C., Bonan, G., Oleson, K. W., et al. (2006). GLACE: The global land-atmosphere coupling experiment. Part II: Analysis. Journal of Hydrometeorology, 7, 611–625. https://doi.org/10.1175/JHM511.1
- Haghighi, E., Short Gianotti, D. J., Akbar, R., Salvucci, G. D., & Entekhabi, D. (2018). Soil and atmospheric controls on the land surface energy balance: A generalized framework for distinguishing moisture-limited and energy-limited evaporation regimes. *Water Resources Research*, 51, 1831–1851. https://doi.org/10.1002/2015WR017464
- Hirschi, M., Seneviratne, S. I., Alexandrov, V., Boberg, F., Boroneant, C., Christensen, O. B., et al. (2011). Observational evidence for soilmoisture impact on hot extremes in southeastern Europe. *Nature Geoscience*, 4, 17–21. https://doi.org/10.1038/ngeo1032
- Holmes, T. R. H., Crow, W. T., & Hain, C. (2013). Spatial patterns in timing of the diurnal temperature cycle. Hydrology and Earth System Sciences, 17, 3695–3706. https://doi.org/10.5194/hess-17-3695-2013
- Holmes, T. R. H., Crow, W. T., Hain, C., Anderson, M. C., & Kustas, W. P. (2015). Remote sensing of environment diurnal temperature cycle as observed by thermal infrared and microwave radiometers. *Remote Sensing of Environment*, 158, 110–125. https://doi.org/10.1016/j. rse.2014.10.031

Huffman, G., 2015. GPM Level 3 IMERG Final Run Half Hourly 0.1 × 0.1 Degree Precipitation, version 05 (Goddard Space Flight Center Distributed Active Archive Center (GSFC DAAC), 2015).

Idso, S. B., Schmugge, T. J., Jackson, R. D., & Reginato, R. J. (1975). Remote sensing of surface soil water status. Journal of Geophysical Research, 80(21), 3044–3049. https://doi.org/10.1029/JC080i021p03044

- James, G., Witten, D., Hastie, T., & Tibshirani, R. (2014). An introduction to statistical learning: With Applications in R. New York: Springer.
- Jarque, C. M., & Bera, A. K. (1980). Efficient test for normality, homoscedasticity and serial independence of regression residuals. *Economics Letters*, 6(3), 255–259. https://doi.org/10.1016/0165-1765(80)90024-5
- Kerr, Y., Waldteufel, P., Wigneron, J.-P., Delwart, S., Cabot, F., Boutin, J., et al. (2010). The SMOS mission: New tool for monitoring key elements of the global water cycle. *Proceedings of the IEEE*, *98*(5), 666–687. https://doi.org/10.1109/JPROC.2010.2043032

Kim, S., 2013. Ancillary data report: Landcover classification. Jet Propuls. Lab. Calif. Inst. Technol., JPL D-53057.

Konings, A. G., & Gentine, P. (2017). Global variations in ecosystem-scale isohydricity. Global Change Biology, 23(2), 891–905. https://doi. org/10.1111/gcb.13389

- Konings, A. G., Piles, M., Rotzer, K., McColl, K. A., Chan, S. K., & Entekhabi, D. (2016). Vegetation optical depth and scattering albedo retrieval using time series of dual-polarized L-band radiometer observations. *Remote Sensing of Environment*, 172, 178–189. https://doi. org/10.1016/j.rse.2015.11.009
- Koster, R. D., Dirmeyer, P. A., Guo, Z., Bonan, G., Chan, E., Cox, P., et al. (2004). Regions of strong coupling between soil moisture and precipitation. *Science*, *305*, 1138–1140. https://doi.org/10.1126/science.1100217
- Koster, R. D., Guo, Z., Dirmeyer, P. A., Bonan, G., Chan, E., Cox, P., et al. (2006). GLACE: The global land-atmosphere coupling experiment. Part I: Overview. Journal of Hydrometeorology, 7, 590–610. https://doi.org/10.1016/S0022-1694(00)00237-7.Inundation
- Koster, R. D., Schubert, S. D., & Suarez, M. J. (2009). Analyzing the concurrence of meteorological droughts and warm periods, with implications for the determination of evaporative regime. *Journal of Climate*, *22*, 3331–3341. https://doi.org/10.1175/2008JCL12718.1

Lakshmi, V., Jackson, T. J., & Zehrfuhs, D. (2003). Soil moisture-temperature relationships: Results from two field experiments. *Hydrological Processes*, 17, 3041–3057. https://doi.org/10.1002/hyp.1275

Lin, Y. S., Medlyn, B. E., Duursma, R. A., Prentice, I. C., Wang, H., Baig, S., et al. (2015). Optimal stomatal behaviour around the world. *Nature Climate Change*, 5, 459–464. https://doi.org/10.1038/nclimate2550

- Mann, H. B., & Whitney, D. R. (1947). On a test of whether one of two random variables is stochastically larger than the other. Annals of Mathematical Statistics, 18(1), 50–60. https://doi.org/10.1214/aoms/1177730491
- Massey, F. J. (1951). The Kolmogorov-Smirnov test for goodness of fit. Journal of the American Statistical Association, 46, 66–78. https://doi. org/10.1198/016214505000000817

Mathon, V., & Laurent, H. (2001). Life cycle of Sahelian mesoscale convective cloud systems. Quarterly Journal of the Royal Meteorological Society, 127(572), 377–406. https://doi.org/10.1002/qj.49712757208

- McColl, K. A., He, Q., Lu, H., & Entekhabi, D. (2019). Short-term and long-term surface soil moisture memory time scales are spatially anticorrelated at global scales. *Journal of Hydrometeorology*, 20, 1165–1182. https://doi.org/10.1175/jhm-d-18-0141.1
- McColl, K. A., Wang, W., Peng, B., Akbar, R., Short Gianotti, D. J., Lu, H., et al. (2017). Global characterization of surface soil moisture drydowns. *Geophysical Research Letters*, 44, 3682–3690. https://doi.org/10.1002/2017GL072819
- Miralles, D. G., Gentine, P., Seneviratne, S. I., & Teuling, A. J. (2019). Land-atmospheric feedbacks during droughts and heatwaves: State of the science and current challenges. *Annals of the New York Academy of Sciences*, 1436(1), 19–35. https://doi.org/10.1111/nyas.13912
 Miralles, D. G., Van, Der Pere, M. J. Terling, A. L. & De Jun, P. A. M. (2012). Geil meisture temperature coupling. A multicula characteristic science and current characteristic science

Miralles, D. G., Van Den Berg, M. J., Teuling, A. J., & De Jeu, R. A. M. (2012). Soil moisture-temperature coupling: A multiscale observational analysis. *Geophysical Research Letters*, 39, L21707. https://doi.org/10.1029/2012GL053703

Panwar, A., Kleidon, A., & Renner, M. (2019). Do surface and air temperatures contain similar imprints of evaporative conditions? Geophysical Research Letters, 46, 3802–3809. https://doi.org/10.1029/2019GL082248 Peng, L., Zeng, Z., Wei, Z., Chen, A., Wood, E. F., & Sheffield, J. (2019). Determinants of the ratio of actual to potential evapotranspiration. *Global Change Biology*, 25(4), 1326–1343. https://doi.org/10.1111/gcb.14577

Rodriguez-Iturbe, I., Entekhabi, D., & Bras, R. L. (1991). Nonlinear dynamics of soil moisture at climate scales 1. Stochastic analysis. Water Resources Research, 27(8), 1899–1906. https://doi.org/10.1029/91WR01035

Roundy, J. K., Ferguson, C. R., & Wood, E. F. (2012). Temporal variability of land-atmosphere coupling and its implications for drought over the Southeast United States. *Journal of Hydrometeorology*, 14, 622–635. https://doi.org/10.1175/JHM-D-12-090.1

Santanello, J. A., Dirmeyer, P. A., Ferguson, C. R., Findell, K. L., Tawfik, A. B., Berg, A., et al. (2018). Land-atmosphere interactions the LoCo perspective. *Bulletin of the American Meteorological Society*, 99, 1253–1272. https://doi.org/10.1175/BAMS-D-17-0001.1

Santanello, J. A., & Friedl, M. A. (2003). Diurnal covariation in soil heat flux and net radiation. *Journal of Applied Meteorology*, 42, 851–862. https://doi.org/10.1175/1520-0450(2003)042<0851:dcishf>2.0.co;2

Scholes, R. J., Gureja, N., Giannecchinni, M., Dovie, D., Wilson, B., Davidson, N., et al. (2001). The environment and vegetation of the flux measurement site near Skukuza, Kruger National Park. *Koedoe*, 44, 73–84.

Schwingshackl, C., Hirschi, M., & Seneviratne, S. I. (2017). Quantifying spatiotemporal variations of soil moisture control on surface energy balance and near-surface air temperature. *Journal of Climate*, 30, 7105–7124. https://doi.org/10.1175/JCLI-D-16-0727.1

Seneviratne, S. I., Corti, T., Davin, E. L., Hirschi, M., Jaeger, E. B., Lehner, I., et al. (2010). Earth-science reviews investigating soil moisture– climate interactions in a changing climate: A review. *Earth-Science Reviews*, 99, 125–161. https://doi.org/10.1016/j.earscirev.2010.02.004

Shellito, P. J., Small, E. E., & Livneh, B. (2018). Controls on surface soil drying rates observed by SMAP and simulated by the Noah land surface model. *Hydrology and Earth System Sciences*, 22, 1649–1663. https://doi.org/10.5194/hess-22-1649-2018

Shinoda, M., & Yamaguchi, Y. (2003). Influence of soil moisture anomaly on temperature in the Sahel: A comparison between wet and dry decades. *Journal of Hydrometeorology*, *4*, 437-447. https://doi.org/10.1175/1525-7541(2003)4<437:IOSMAO>2.0.CO;2

Short Gianotti, D. J., Rigden, A. J., Salvucci, G. D., & Entekhabi, D. (2019). Satellite and station observations demonstrate water availability's effect on continental-scale evaporative and photosynthetic land surface dynamics. *Water Resources Research*, 55, 540–554. https://doi.org/10.1029/2018WR023726

Taylor, C. M., De Jeu, R. A. M., Harris, P. P., Dorigo, W. A., & Africa, W. (2012). Afternoon rain more likely over drier soils. *Nature*, 489, 423–426. https://doi.org/10.1038/nature11377

Taylor, C. M., Gounou, A., Guichard, F., Harris, P. P., Ellis, R. J., Couvreux, F., & de Kauwe, M. (2011). Frequency of Sahelian storm initiation enhanced over mesoscale soil-moisture patterns. *Nature Geoscience*, *4*, 4–7. https://doi.org/10.1038/NGEO1173

Teuling, A. J., Seneviratne, S. I., Williams, C., & Troch, P. A. (2006). Observed timescales of evapotranspiration response to soil moisture. Geophysical Research Letters, 33, L23403. https://doi.org/10.1029/2006GL028178

Trigo, I. F., Dacamara, C. C., Viterbo, P., Roujean, J. L., Olesen, F., Barroso, C., et al. (2011). The satellite application facility on land surface analysis. *International Journal of Remote Sensing*, 32(10), 2725–2744. https://doi.org/10.1080/01431161003743199

Trigo, I. F., Monteiro, I. T., Olesen, F., & Kabsch, E. (2008). An assessment of remotely sensed land surface temperature. Journal of Geophysical Research, 113, D17108. https://doi.org/10.1029/2008JD010035

Tuttle, S., & Salvucci, G. (2016). Empirical evidence of contrasting soil moisutre-precipitation feedbacks across the United States. Science, 352(6287), 825–828. https://doi.org/10.1126/science.aaa7185

Tuttle, S. E., & Salvucci, G. D. (2017). Confounding factors in determining causal soil moisture-precipitation feedback. Water Resources Research, 5531–5544. https://doi.org/10.1002/2016WR019869

Williams, I. N., & Torn, M. S. (2015). Vegetation controls on surface heat flux partitioning, and land-atmosphere coupling. *Geophysical Research Letters*, 42, 9416–9424. https://doi.org/10.1002/2015GL066305

Zscheischler, J., Orth, R., & Seneviratne, S. I. (2015). A sub-monthly database for detecting changes in vegetation-atmosphere coupling. Geophysical Research Letters, 42, 9816–9824. https://doi.org/10.1002/2015GL066563.Received